Author's response to reviews

Title: Intravenous immunoglobulin in the treatment of primary trigeminal neuralgia refractory to carbamazepine: a study protocol (ISRCTN33042138)

Authors:

Andreas Goebel (andreasgoebel@rocketmail.com)
Andrew Moore (andrew.moore@pain-relief-unit.oxford.ac.uk)
Rosamund Weatherall (R.Weatherall@icrf.icnet.uk)
Norbert Roewer (nroewer@anaesthesie.uni-wuerzburg.de)
Robert Schedel (rschedel@anaesthesie.uni-wuerzburg.de)
Guenter Sprotte (gsprotte@anaesthesie.uni-wuerzburg.de)

Version: 2 Date: 15 Oct 2002

As requested, a title page and abstract are now preceding the main manuscript.